Author’s response to reviews

Title: Type B insulin resistance syndrome with Scleroderma successfully treated with multiple immune suppressants after eradication of Helicobacter pylori infection: a case report

Authors:

Guo-Qing Yang (endocrine301@126.com)
Yi-Jun Li (nmyxylyjy@126.com)
Jing-Tao Dou (jingtaodou@163.com)
Bao-An Wang (wangbaoan11@126.com)
Ju-Ming Lu (lujuming301@126.com)
Yi-Ming Mu (muyiming@301hospital.com.cn)

Version: 2 Date: 17 Mar 2016

Author’s response to reviews:

Dear Mr. Shipley,

It’s our pleasure to hear that you consider our manuscript potentially acceptable by BMC Endocrine Disorders. We appreciate the comments from the reviewers and editors and we have addressed them in a point-by-point manner.

We hope the current version is publishable in BMC Endocrine Disorders.

Best regards

Yiming Mu

Point-by-point response

Reviewer reports:

After the main sections in the body of the manuscript, all article types must contain the following items under ‘Declarations’:

* Ethics and Consent to Participate statement
* Consent to Publish statements
* Competing Interest statement
* Au Contributions (if there is more than one author)
* Availability of data and materials statement
* Abbreviations (if present)
* Notes (if present)

Reply: we have revised section of “Declarations” according to the above requirements.

--------------------

Editorial Requests

--------------------

Please note that all submissions to BMC Endocrine Disorders must comply with our editorial policies. Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements this will cause a delay whilst the issue is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

Ethics:

If your study involves humans, human data or animals, then your article should contain an ethics statement which includes the name of the committee that approved your study.

If ethics was not required for your study, then this should be clearly stated and a rationale provided.

Consent:

If your article is a prospective study involving human participants then your article should include a statement detailing consent for participation.

If individual clinical data is presented in your article, then you must clarify whether consent for publication of these data was obtained.

Availability of supporting data:

BioMed Central strongly encourages all data sets on which the conclusions of the paper rely be either deposited in publicly available repositories (where available and appropriate) or presented in the main papers or additional supporting files, in machine-readable format whenever possible. Authors must include an Availability of Data and
Materials section in their article detailing where the data supporting their findings can be found. The Accession Numbers of any nucleic acid sequences, protein sequences or atomic coordinates cited in the manuscript must be provided and include the corresponding database name.

Authors Contributions:

Your 'Authors Contributions' section must detail the individual contribution for each individual author listed on your manuscript.

Reply: we believe the current manuscript complies with the editorial policies of BMC Endocrine Disorders.